Author's response to reviews

Title: Fasudil improves survival and promotes skeletal muscle development in a mouse model of spinal muscular atrophy

Authors:

Melissa Bowerman (melissabowerman@hotmail.com)
Lyndsay M Murray (lmmurray82@googlemail.com)
Justin G Boyer (jusboyer@hotmail.com)
Carrie L Anderson (carrie.l.anderson@gmail.com)
Rashmi Kothary (rkothary@ohri.ca)

Version: 8 Date: 10 February 2012

Author's response to reviews:

Dear BMC Series Editorial team,

In the present revision, we have have changed reference number 38 for number 17, because they were duplicated. Both the main text and reference list have been corrected and updated.

Regards,
Melissa Bowerman